Author’s response to reviews

Title: Listening to women's voices: the experience of giving birth with paramedic care in Queensland, Australia

Authors:
Belinda Flanagan (BFlanaga@usc.edu.au)
Bill Lord (wlord@usc.edu.au)
Rachel Reed (reel1@usc.edu.au)
Gail Crimmins (gcrimmin@usc.edu.au)

Version: 5 Date: 10 Nov 2019

Author’s response to reviews:

Thank you for your feedback - The title has been changed to reflect the editors comments and to differentiate between the larger study and this publication. The reference to the larger study is included in the abstract as per the editors comments.